Author's response to reviews

Title: Serum concentration of eicosapentaenoic acid is associated with cognitive function in patients with coronary artery disease

Authors:

Shusuke Yagi (syagi@tokushima-u.ac.jp)
Tomoya Hara (tomoyahara@tokushima-u.ac.jp)
Rie Ueno (linda_s610105@yahoo.co.jp)
Ken-ichi Aihara (aihara@tokushima-u.ac.jp)
Daiju Fukuda (daiju.fukuda@tokushima-u.ac.jp)
Akira Takashima (takashima-a@tokushima-u.ac.jp)
Junko Hotchi (jhotchi@tokushima-u.ac.jp)
Takayuki Ise (isetaka@tokushima-u.ac.jp)
Koji Yamaguchi (yamakoji3@tokushima-u.ac.jp)
Takeshi Tobiume (tobiume.takeshi@tokushima-u.ac.jp)
Takashi Iwase (iwasetaka@tokushima-u.ac.jp)
Hirotsugu Yamada (yamadah@tokushima-u.ac.jp)
Takeshi Soeki (soeki@tokushima-u.ac.jp)
Tetsuzo Wakatsuki (wakatsukitz@tokushima-u.ac.jp)
Michio Shimabukuro (mshimabukuro-ur@umin.ac.jp)
Masashi Akaike (akaike.masashi@tokushima-u.ac.jp)
Masataka Sata (masataka.sata@tokushima-u.ac.jp)

Version: 3 Date: 28 October 2014

Author’s response to reviews: see over
We thank the reviewers for their valuable comments and suggestions.

Comments by the reviewers are below, followed by our responses.

Referee 1:
The authors have thoroughly addressed my comments in their revision of the manuscript. I noted only one very minor revision remaining: in Supplemental Table 2, “DHLA” should be revised to “DGLA.”

As the reviewer pointed out, we changed DHLA to DGLA in Supplemental Table 2. We appreciate your comments and suggestions to improve the manuscript.

Referee 2:

I think this article has been properly modified according to my comments enough to be published on NUTRITION JOURNAL.

We appreciate your comments and suggestions to improve the manuscript.